Author's response to reviews

Title: Human immunodeficiency virus infection and cerebral malaria in children in Uganda: a case-control study

Authors:

Peace D Imani (imani9p@yahoo.com)
Philippa Musoke (pmusoke@mujhu.org)
Justus Byarugaba (byarugabai@yahoo.com)
James K Tumwine (kabaleimc@gmail.com)

Version: 10 Date: 13 January 2011

Author's response to reviews:

Major revisions
Abstract: in the submission system matched with that in the manuscript.
Each affiliation written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country.

Minor revisions
Tables reformatted and merged cells removed.
The figure has been revised.
Checked for any typographical errors and final corrections made.